Author’s response to reviews

Title: WhatsApp embedded in routine service delivery for smoking cessation: Effects on abstinence rates in a randomized controlled study

Authors:

Seyfi Durmaz (seyfidurmaz@gmail.com)

Isil Ergin (isil.ergin@ege.edu.tr)

Raika Durusoy (raika.durusoy@ege.edu.tr)

Hur Hassoy (hur.hassoy@ege.edu.tr)

Ayhan Caliskan (ayhan.caliskan@ege.edu.tr)

Pinar Okyay (pinarokyay@adu.edu.tr)

Version: 3 Date: 20 Mar 2019

Author’s response to reviews:

Dear Editor,

We are grateful for your time and help.

Here are our responses to the requested corrections on the manuscript:

1. Trial Registration:

"In the Trial registration section at the end of the Abstract, Please remove the information relating to ethics approval. Please only list the trial registry and the unique identifying number, as well as the date of registration. Please note that there should be no space between the letters and numbers of the trial registration number. If registration took place after the first participant was enrolled, please state also “Retrospectively registered” at the end of this section"
The information relating to ethics approval is removed. We added “Retrospectively registered” to the related sentence. The final version of the sentence is “Trial registration: This trial is retrospectively registered online at ClinicalTrials.gov with the identifier NCT03714971.”

2. Formatting:

"a) Please amend the following headings in your manuscript, as indicated:
- “Introduction”: please change this to “Background”

b) Please move your 'Limitations' Section so that it forms part of the Discussion and comes before the Conclusions"

a) The heading “Introduction” is changed to “Background”

b) 'Limitations' Section is moved in to the Discussion part before the Conclusions.

3. Image attribution:

"Please state in the cover letter whether the cartoon images, e.g. money, hands, and bodies, depicted in figure 1 are your own or taken from another source.

If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it.

If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time."
Should you require an alternative source you may wish to try Wikimedia Commons: http://commons.wikimedia.org/wiki/Main_Page

The image is removed and the Figures are renumbered (also in the text)

4. Clean Manuscript

"Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files."

The entire manuscript is checked and the clean version is formed

Kindest Regards

Dr. Seyfi Durmaz